Author's response to reviews

Title: Prevalence of dyslipidemia and its association with insomnia in a community based population in China

Authors:

Yiqiang Zhan (zhanyiqiang@gmail.com)
Fen Zhang (w524zf@163.com)
Jinsong Wang (driinsonqw@gmail.com)
Yihong Sun (dryihongs@gmail.com)
Rongjing Ding (drrongjiingd@hotmail.com)
Dayi Hu (drdayihu@hotmail.com)
Jinming Yu (jmyu64@gmail.com)

Version: 2
Date: 9 June 2014

Author's response to reviews:

Dear Editor,

The line and page numbers were added to the manuscript now.

Best,

Yiqiang Zhan